Author's response to reviews

Title: Changing the prognosis from hopeless to good after advanced and comprehensive periodontal treatment of aggressive periodontitis: a case report.

Authors:

Wisam Kamil (wissamalh9@gmail.com)
Lina Al Bayati (lenahwm@yahoo.co.uk)
Akbar S. Hussin (ohara6161@yahoo.com.au)
Haszelini Hassan (haszel_2904@rocketmail.com)

Version: 2
Date: 18 May 2015

Author's response to reviews: see over
Dear Editor-in-Chief

Please find enclosed our manuscript (Changing the prognosis from hopeless to good after advanced and comprehensive periodontal treatment of aggressive periodontitis: a case report) by kamil W et al., which we would like to submit for publication as a case report in Journal of Medical Case Reports.

To our knowledge, this is the first report that shows a comprehensive step-by-step approach in the treatment of a challenged case in the aesthetic zone of an aggressive periodontitis patient through managing a circumferential bony defects associated with two upper central incisors. We believe our report would be interested to the readership of Journal of Medical Case Reports as it demonstrates to periodontists the approach to achieve a favorable long-term prognosis in order to preserve natural healthy dentition with support and aesthetic, rather than the option of tooth extraction and replacement by removable partial appliances.

Please address all correspondence to:
Assistant Professor Dr. Wisam A. Kamil
BDS, MSc Dental Surgery (Periodontology)
Periodontics Unit,
Faculty of Dentistry,
International Islamic University Malaysia.
Bandar Indera Mahkota,
25200 Kuantan, Pahang D/M, Malaysia.
H/P: +60199724947
Office: +6095705506
Email: wissamalh9@gmail.com

We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal.
All authors have approved the manuscript and agree with its submission to Journal of Medical Case Reports and declare that they have no competing interests.

Thank you for having the time to consider this submission and look forward to hearing from you in the near future.

Sincerely,
Wisam